Author's response to reviews

Title: Uptake of HIV counselling and testing among pregnant women at different levels of health facilities - experiences from a community-based study in Northern Vietnam.

Authors:

Hanh Nguyen (n_thuyhanh@yahoo.com)
Tine Gammeltoft (tine.gammeltoft@anthro.ku.dk)
Vibeke Rasch (vira@sund.ku.dk)

Version: 4 Date: 29 March 2010

Author's response to reviews: see over
Dear Editors,

First of all, we would like to thank the Editor for making process on our paper. I have opportunity to look at again the data and the paper that I resubmitted to you on 8th of March, and I recognized some mistake of data. I would like to resubmit my paper with the correction of data and the changes of related text of the paper. I would like you to pay attention and I am so sorry for your inconvenience. Following are the list of changes (I marked the yellow of what I have changed)

1. On page 1 of the Abstract, line 6, the first paragraph of Results, we have revised the text “(aOR = 44, CI: 19-101)”

2. On page 6, in Results section, under sub-heading “Early testing and provision of counseling”, the first paragraph, line 4-5, we have change the text: “with OR = 43 (CI: 19-98). After adjusting for “time of fist ANC visit”, the OR was the almost the same (OR: 44, CI: 19 – 101.”

3. On page 7, in discussion section, the forth paragraph, line 12-13-14, we have change the text: “the result was not significantly changed.”

4. On page 14, table 1, at the last variable: HIV testing by gestational week, we have correct data.

5. On page 16, table 3, we have correct all data.

Thanks for attention and hope that our paper will be considered for publication.
With the best regards,
Vibeke Rasch, Tine Gammeltoft and Nguyen Thi Thuy Hanh